Author’s response to reviews

Title: Multimorbidity Patterns and Health Service Use in Swedish 85 year olds: An Exploratory Study

Authors:

Huan-Ji Dong (huanji.dong@liu.se)
Ewa Wressle (ewa.wressle@lio.se)
Jan Marcusson (jan.marcusson@liu.se)

Version: 5 Date: 29 June 2013

Author’s response to reviews: see over
29 Jun 2013
Ms Catherine Olino
BMC Geriatrics Editorial team

Dear Ms Catherine Olino!

MS: 2009248583926952
Multimorbidity Patterns and Health Service Use in Swedish 85 year olds: An Exploratory Study (Huan-Ji Dong, Ewa Wressle and Jan Marcusson)

As requested (2013-06-26), we would like to resubmit a revised version of our paper with highlight/track changes. In this new version, we highlight the modified text (major changes) and tables with yellow colour.

We look forward to hearing from you regarding your comments and decision on this revised paper.

Yours sincerely,
Huan-Ji Dong